Author's response to reviews

Title: Does clinical examination aid in the diagnosis of urinary tract infections in women? A systematic review and meta-analysis.

Authors:

David Medina-Bombardó (dbombardo@gmail.com)
Antoni Jover-Palmer (ajover@ibsalut.caib.es)

Version: 8 Date: 25 September 2011

Author's response to reviews: see over
Dear Editor:

We have proceeded to change the last format recommendations for the manuscript.

Major revision
- References: Please ensure that references are cited in numerical order. Done! We have detected only one mistake in this concept: references 8 and 9 were disordered.
- Tables: Please ensure that table one is placed in both columns and rows, or include the information within the manuscript text - you can use indentation to highlight the text. Done! The table one are been restructured in rows and columns.

Minor revisions
- Tables: Please remove the tables' legends from within the table cells. Place the legends below their respective tables. Done! All legends are under and outside their table.
- Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. Done!

Sincerely

David Medina-Bombardó
Antoni J. Jover-Palmer